Author’s response to reviews

Title: Clinicopathological variables influencing overall survival, recurrence and post-recurrence survival in resected stage I non-small-cell lung cancer

Authors:

Chengdi Wang (wangchengdi2020@126.com)
Yuxuan Wu (allformayday@sina.com)
Jun Shao (shaojun1997@foxmail.com)
Dan Liu (liudanscu@qq.com)
Weimin Li (weimi003@scu.edu.cn)

Version: 2 Date: 06 Feb 2020

Author’s response to reviews:

We thank the editor for the constructive criticisms again that have helped us to improve our manuscript. The point-by-point response to the comments is given below and the changes occur in the revised manuscript.

1. Please include the full name of the ethics committee (and the institute to which it belongs) that approved the study and the committee’s reference number if appropriate.

Answer: The full name of the ethics committee is “The Institutional Review Board of West China Hospital, Sichuan University” that approved the retrospective study (Ref no. #2018–270) in which informed consent was waived, but patient confidentiality was protected.

2. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Answer: The funding bodies had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.
Answer: We have checked and made sure that the manuscript was uploaded as a single, final, clean version, and all figures, tables were cited within the text according to the requirements.